NHS National Institute for Health Research

NETSCC, HTA

13 November 2012

The Health Technology Assessment programme is managed by NETSCC, HTA as part of the NIHR Evaluation, Trials and Studies Coordinating Centre at the University of Southampton.

Alpha House, University of Southampton Science Park Southampton SO16 7NS

Suggest a topic for research via our online form at www.hta.ac.uk/suggest

tel: +44(0)23 8059 5586 fax: +44(0)23 8059 5639 email: hta@hta.ac.uk



The NIHR Evaluation, Trials and Studies Coordinating Centre (NETSCC), based at the University of Southampton, manages evaluation research programmes and activities for the NIHR

Health Technology Assessment Programmetel: +44(0)23 8059 5586email: hta@hta.ac.ukNational Institute for Health ResearchEvaluation, Trials and Studies Coordinating Centreweb: www.hta.ac.ukUniversity of Southampton, Alpha Housefax: +44(0)23 8059 5639web: www.hta.ac.ukEnterprise Road, Southampton, SO16 7NSweb: www.hta.ac.uk

REVISED PROTOCOL 02/09/2012

Protocol NIHR HTA Programme project number 11/130/02

1. Research question

What is the risk of adverse events from aspirin, taken for prophylactic use for the primary prevention of cardiovascular disease or cancer? Analysis using randomised controlled trials (RCTs), systematic reviews and meta-analyses from RCTs.

2. Name of TAR team and project 'lead'

Produced by:	Warwick Evidence
	Health Sciences Research Institute
	Medical School
	University of Warwick
	Coventry
	CV4 7AL
Lead Author:	Paul Sutcliffe
Co-authors:	Martin Connock
	Tara Gurung
	Kandala Ngianga-Bakwin
	Samantha Johnson
	Amy Grove
	Aileen Clarke
Correspondence to:	Dr Paul Sutcliffe, Warwick Evidence, Warwick Medical School,
	University of Warwick, Coventry, CV4 7AL
Tel:	02476 150189
Fax:	02476 528375
Email:	p.a.sutcliffe@warwick.ac.uk
Date Completed:	31 August 2012

3. Plain English Summary

Taken in appropriate dosage, long term use of aspirin is thought to protect people from future heart problems and cancer. However, for some individuals, taking aspirin has unwanted side effects such as bleeding and stomach pain. Therefore, potential benefit of protection must be balanced against possible harm from side effects. This balance may be different for different people and it is particularly important to know the risk of side effects from preventative aspirin for those people as yet free from, but at risk of, developing cardiovascular disease or cancer. This report aims to find the current scientific evidence about this and to summarise this literature by looking in detail at systematic reviews and meta-analyses on the occurrence of side effects from the preventative use of aspirin in people free of cardiovascular disease and cancer.

4. Decision problem

Objectives

- i. To identify RCTs, systematic reviews and meta-analyses of RCTs of the prophylactic use of aspirin in the primary prevention of cardiovascular disease or cancer.
- ii. With particular reference to adverse events, undertake an overview and quality assessment of the identified systematic reviews and meta-analyses.
- To undertake study level meta-analysis to investigate the relative influence of individual studies on pooled estimates of risk of adverse events reported in identified systematic reviews and meta-analyses.
- iv. To undertake cumulative meta-analysis on time of study initiation or study publication to investigate influence on pooled estimates of risk of adverse events reported in identified systematic reviews and meta-analyses.
- v. To undertake exploratory multivariable meta-regression of studies in identified systematic reviews and meta-analyses to investigate the potential influence of study level variables on reported pooled estimates of risk of adverse events (e.g. participant age and gender; follow up duration; aspirin dose or dose frequency; level of or type of cardiovascular risk; year of investigation). (Whilst we are aware that it is recommended that each study level variable requires approximately 8 studies, we will emphasise the exploratory nature of the analyses should variables exceed this ratio).
- vi. To summarise, synthesise and assess the recommendations provided in the systematic reviews and meta-analyses reporting on adverse events resulting from prophylactic use of aspirin in primary prevention in the light of objectives i-v and if appropriate to make recommendations for further investigation

4.1 Background

Aspirin (acetylsalicylic acid) is a widely used antiplatelet drug for primary and secondary prevention of cardiovascular events.¹ Typical doses employed range from75 to 325 mg daily or

every other day. Some authors² have defined low dose, medium and high doses of aspirin, but such classification is somewhat arbitrary and subjective.

The regular use of even low dose aspirin appears to increase the risk of death from GI bleeding, cerebrovascular bleeding^{3,4} and may exaggerate the severity of asthma attacks.⁵ Some evidence suggests that relative to higher doses, lower doses may be protective while resulting in fewer adverse effects.⁶ Aspirin related GI bleeding may be more common in older patients (>70 years) and in those with a past history of peptic ulcer.⁷ Discontinuation of long term use has been linked to increased risk of non-fatal myocardial infarction compared to those who continued treatment.⁸

Several guidelines exist that consider the prophylactic use of aspirin; these are based on an assessment of the balance between cardiovascular benefits (e.g. reduced MI and stroke) and various harms (especially bleeding); some recommend widespread employment of aspirin for individuals at increased risk of CVD.^{9,10} Recently, opinion and evidence appear to have shifted. Firstly, benefits in primary prevention of CVD are now generally viewed as relatively modest, remain statistically uncertain, and are an order of magnitude less than that observed in secondary prevention with aspirin, while harms (especially bleeding) occur at relatively high frequency (very high frequency in some populations). Secondly, investigations that use a mix of individual patient data (IPD) and study level meta-analyses of randomised controlled trials (RCTs) and of observational studies, now point to a possible protection against several cancers¹¹ (notably colon cancer). Apparent protection emanates after about five years of aspirin use, and there is also evidence for protection against cancer metastasis. These latter studies have been viewed with caution by some because data from the two largest CVD primary prevention trials were excluded.^{12,13}

4.2 Scoping searches

In November 2011 Warwick Evidence carried out search of current relevant research related to potential harms from aspirin given in low dose (taken as <300mg) for any indication. The aim of the scoping searches was to generate a rapid overview of evidence on the potential harms from prophylactic aspirin (<300mg) for any indication, and to gauge the current status of policy concerning aspirin prophylaxis in primary prevention. Details are provided in *Appendix 1*

Amore recent scoping search (April 2012) undertaken in response to correspondence with NIHR HTA focused on the use of aspirin for primary prevention. This revealed that evidence relating to benefits and risks of prophylactic aspirin is currently a very active area of systematic review and meta-analysis. There are already several recent systematic reviews of prophylactic aspirin for the primary prevention of cardiovascular events,¹⁴⁻¹⁶ each of which have meta-analysed the same basic core of nine randomised controlled trials of primary prevention.¹⁷⁻²⁵ These RCTs have included over 100,000 patients.

Similarly, scoping has indicated the existence of a growing number of reviews and meta-analyses that focus on the possible protection of long term aspirin against cancers and cancer metastasis. Primary prevention RCTs, secondary prevention studies,²⁶ and observational studies have featured in these analyses and, in some, IPD meta-analyses¹¹ have been conducted. In general it appears that adverse events (e.g. bleeding) are rarely reported in these cancer protection studies, except where studies have been included from amongst the core nine RCTs of long term aspirin for primary prevention cardiovascular disease.

In summary: The RCT evidence-base to address the protocol research questions does not appear to have grown since the publication of the AAA trial in 2011¹⁹ (several unreported on-going trials have been identified in scoping). This evidence has been subject to intense systematic review and meta-analysis including many study level meta-analytic investigations and a landmark IPD meta-analysis published in 2009.¹ In general the published meta-analyses appear to be well conducted and current according to the time they were undertaken; however inferences and conclusions appear to differ from study to study. Thus far it appears that no overview of these meta-analyses and reviews has been undertaken or published.

We therefore plan:

A] With particular reference to the occurrence of adverse events, to undertake an overview of the systematic reviews and meta-analyses of RCTs which have investigated the long-term use of aspirin for primary prevention of CVD or cancer.

B] So as to identify changes through time, undertake cumulative meta-analysis of these RCTs.

C] So as to investigate the relative influence of individual RCTs on pooled estimates, undertake study level meta-analysis of the RCTs.

D] So as to identify study level variable that influence the occurrence of adverse events undertake exploratory multi variable meta-regression of the RCTs.

These options are relatively straightforward to undertake. Option A] is justified on the grounds that although a plethora of meta-analytic studies have been generated, no overview has yet been published that compares them, particularly with regard to adverse events, or sets them in context. Options B] to D] are justified since they can address how aspirin use in the primary prevention of CVD or cancer has evolved since clinical trials in the 1980s, and the introduction of guidelines on the use of aspirin in primary prevention from trials published up to 2010. Moreover, trials' conditions and patients' characteristics have also evolved over time and there is considerable heterogeneity among randomized trials. In the meanwhile, preventative treatments for CVD have greatly changed (introduction of statins and anti-hypertensive drugs), and there are observed differences in the outcomes from the trials. Therefore, early results cannot be easily compared with later studies, a limitation that prior meta-analyses accounted for only partially or not at all.

Alternative avenues of investigation have been considered but not judged viable on reviewer's advice, and on the basis of the project's time scale and remit from NIHR HTA. These are as follows: (i) to expand the analysis so as to include observational studies. Since RCTs account for over 100,000 patients and the ratio of RCTs to cohort studies in a previous meta-analysis that was restricted to patients with diabetes was about 4:1, including the results from such studies may not add significant value to knowledge already accumulated; (ii) to perform IPD meta-analysis of RCTs, by expanding on the six primary prevention RCTs previously analysed by Baigent et al 2009¹ Negotiating agreement for access to RCT data would be difficult and time consuming and possibly unsuccessful since it is very likely Baigent et al. requested IPD for these studies but were unable to obtain it. Because of the low probably of obtaining IPD and the time required to obtain and analyse it, this option was not judged viable within the project time scale and remit. (iii) Expand the analysis to include IPD from the THIN registry (a UK NHS general practice registry that holds data on 3 million patients, about 2,000 of whom were prescribed low dose aspirin). An industry sponsored analysis of GI bleeding resulting from use of low dose aspirin has already been published using data in the THIN registry. Analysis of intracranial bleeding would probably be hampered by lack of discrimination between types of stroke entered into the registry. Furthermore the larger number of participants in the available RCTs brings into question the added value from such an undertaking.

5. **Report methods for synthesis of clinical evidence**

With particular reference to adverse events an overview will be undertaken of RCTs, systematic reviews and meta-analyses of RCTs of the prophylactic use of aspirin for the primary prevention of cardiovascular disease or cancer published since 2008. The general principles recommended by NHS Centre for Reviews and Dissemination (CRD) will be applied.²⁷

5.1 Identification and selection of studies

Scoping searches were undertaken to assess the volume and type of literature relating to the assessment question. A search strategy will be developed which focuses the searches to meet the inclusion and exclusion criteria (see below). All searches will be undertaken in September2012.

5.1.1 Search strategy for clinical effectiveness

An iterative procedure will be used to inform the development of the search strategy, with input from clinical advisors and previous HTA and systematic reviews (e.g. Bartlolucci et al. 2011,¹⁴ Berger et al. 2011,¹⁶ Rothwell et al. 2012¹¹). Copies of search strategies to be used in the major databases are provided in

Appendix 2. These draft search strategies developed for MEDLINE will be adapted as appropriate for other databases. The strategies cover the concepts of aspirin, prevention and control,¹ and selected publication types (systematic reviews, meta-analyses and randomised controlled trials).

The search strategy will comprise the following main elements:

- Searching of electronic bibliographic databases
- Contact with experts in the field
- Scrutiny of references of included studies

Databases will include:

MEDLINE; MEDLINE In-Process & Other Non-Indexed Citations; EMBASE; Cochrane Database of Systematic Reviews; CENTRAL; DARE, NHS EED, HTA databases (NHS-CRD); Science Citation Index and Conference Proceedings (Web of Science); UKCRN Portfolio Database; Clinical Trials.gov.

In addition, the reference lists of relevant articles will be checked and various health services research related resources will be consulted via the Internet. These are likely to include HTA organisations, including the NIHR and the National Research Register (NRR) Archive, guideline producing bodies, generic research and trials registers:

- Medicines and Healthcare products Regulatory Authority (MHRA)
- US Food and Drug Administration
- The Aspirin Foundation
- The British Cardiovascular Society
- European Society of Cardiology
- American Heart Association
- Cancer Research UK
- Institute of Cancer Research
- American Association for Cancer Research

5.1.2 Inclusion of relevant studies

¹MeSH floating sub-heading pc.fs which will be used in MEDLINE and EMBASE. An alternative will be considered for other databases.

Study design:

RCTs, systematic reviews and meta-analyses of RCTs on the use of aspirin in the primary prevention of CVD or cancer.

Studies will be defined as primary prevention if participants with previous ischaemic vascular events or relevant cancers have been excluded (or are separately identified and can be excluded) or represent <20% of included participants.

To be included, systematic reviews needed to report data from studies separately with a minimum of 50% of studies being eligible RCTs.

Population:

Adults aged over 18 years without clinical cardiovascular disease (established or symptomatic), or adults aged over 18 years without cancer (established or symptomatic).

Intervention:

Aspirin (any dosage) taken prophylactically for primary prevention of cancer or cardiovascular disease.

Aspirin combination therapy (e.g. Aspirin combined with a second antithrombotic agent) will only be included if there are separate placebo and aspirin-only treatment groups, in which case the data from these groups only will be included.

Comparator:

Placebo, no aspirin or no other treatment.

Outcomes:

The primary outcome of interest is the risk of adverse events from prophylactic aspirin for primary prevention, compared to placebo, no aspirin or no other treatment.

Other outcomes reported in the included reviews and meta-analyses will be recorded.

5.1.3 Exclusion criteria

- All study designs other than RCTs, systematic reviews or meta-analyses
- Systematic reviews or meta-analyses that only include secondary prevention studies

- Systematic reviews or meta-analyses that only include observational studies
- Studies not in English

5.2 Review methods

A record of all papers rejected at full text stage and reasons for exclusion will be documented. Titles and abstracts of retrieved studies will be examined for inclusion by two reviewers independently. Disagreement will be resolved by consensus.

5.3 Data extraction strategy

The full data will be extracted by one reviewer and checked by a second. Extraction forms for systematic reviews have been developed (see Appendix 3). Any disagreements will be resolved by discussion. Further discrepancies will be resolved with involvement of a third reviewer when necessary. Summary tables will detail information about study design, participant, intervention, comparator and outcomes. In addition we will provide a summary of the findings and authors conclusions.

Data will be extracted to allow quality assessment of the included studies (see below).

5.4 Quality assessment strategy

Quality criteria will be applied independently by two reviewers, with any disagreements resolved by independent assessment by a third reviewer. Included systematic reviews will be quality assessed using the NHS CRD²⁷ checklist for systematic reviews and the Cochrane Risk of Bias tool²⁸ for RCTs (See Appendix 4).

5.5 Methods of analysis/synthesis

A narrative overview and analysis of included systematic reviews and meta-analyses will be undertaken and supplemented with further meta-analysis.

Data from the included studies will be tabulated and summarised. Meta-analyses will be undertaken using random effects models using STATA software (StataCorp 4905 Lakeway Drive College Station, Texas 77845 USA). Following the scoping searches it is considered that a random effects model is likely to be the primary analysis due to the likely differences in patient characteristics and aspirin doses. Particular attention will be focused on the reporting of adverse events (outcome statistic), the range of adverse event definitions employed in the primary studies,

and how discrepant event definitions have been handled when data has been synthesised by metaanalysts.

We anticipate conducting meta-analyses including cumulative meta-analysis of studies to identify changes through time; study level meta-analysis to investigate the relative influence of individual RCTs and exploratory multi variable meta-regression (we are aware that it is recommended that each variable requires approximately 8 RCTs, however we will emphasise the exploratory nature of the analysis should the variables exceed this ratio). Because of clinical heterogeneity a random effects model will be the method of choice, and tau squared will be recorded. We will explore publication bias using methods in the Cochrane handbook (recommended methods for testing funnel plot asymmetry): and statistical heterogeneity beyond that expected through chance would be investigated using I^2 .

6. Expertise in this TAR team

Warwick Evidence is a technology assessment group located within Warwick Medical School. Warwick Evidence brings together experts in clinical and cost effectiveness reviewing, medical statistics, health economics and modelling. The team planned for the work includes: Dr Paul Sutcliffe and Dr Tara Gurung, who are experienced systematic reviewers; Mrs Samantha Johnson, information specialist; Professor Aileen Clarke, Dr Kandala Ngianga-Bakwin provide epidemiological and statistical expertise; Professor Peter Elwood, University of Cardiff, and Professor Martin Underwood and Dr Saverio Stranges, University of Warwick and Dr Wendy Gregory, Clinical Consultant Gastroenterologist will provide methodological and clinical advice; Ms Amy Grove and Ms Sarah morrow will provide project management and reviewing support.

7. Competing interests of authors and advisors

None of the authors have any competing interests.

8. Timetable/milestones

Draft protocol finalised	TBC
Commissioning decision	TBC
Anticipated start date	17 th September 2012
Draft Final report	30 th November 2012

10. Team members' contributions

Research team:	Warwick Evidence
Lead:	Dr Paul Sutcliffe
Title:	Associate Professor
Address:	Health Sciences Research Institute, Warwick Medical School, University of
	Warwick, Coventry CV4 7AL
Tel:	02476 574505
Email:	p.a.sutcliffe@warwick.ac.uk
Contribution:	Co-ordinate review process, protocol development, assessment for eligibility,
	quality assessment of trials, data extraction, data entry, data analysis, and report
	writing
Name:	Dr Martin Connock
Title:	Senior Research Fellow
Address:	Health Sciences Research Institute, Warwick Medical School, University of
	Warwick, Coventry CV4 7AL
Tel:	02476 574940
Email:	m.connock@warwick.ac.uk
Contribution:	Co-ordinate review process, protocol development, assessment for eligibility,
	quality assessment of trials, data extraction, data entry, data analysis, and report
	writing
Name:	Dr Tara Gurung
Title:	Research Fellow
Address:	Health Sciences Research Institute, Warwick Medical School, University of
	Warwick, Coventry CV4 7AL
Tel:	02476 150711
Email:	t.gurung@warwick.ac.uk
Contribution:	Protocol development, assessment for eligibility, quality assessment of trials,
	data extraction, data entry, data analysis, and report writing
Name:	Dr Kandala Ngianga-Bakwin
Title:	Principal Research Fellow
Address:	Health Sciences Research Institute, Warwick Medical School, University of
	Warwick, Coventry CV4 7AL

Tel:	02476 575054
Email:	N-B.Kandala@warwick.ac.uk
Contribution:	Data entry, data analysis, and statistical modeller
Name:	Mrs Samantha Johnson
Title:	Information Specialist
Address:	The University Library, University of Warwick, Coventry CV4 7AL
Tel:	02476 522427
Email:	Samantha.A.Johnson@warwick.ac.uk
Contribution:	Protocol development, develop search strategy and undertake the electronic
	literature searches
Name:	Ms Amy Grove
Title:	Project Manager
Address:	Health Sciences Research Institute, Warwick Medical School, University of
	Warwick, Coventry CV4 7AL
Tel:	02476 528375
Email:	A.L.Grove@warwick.ac.uk
Contribution:	Retrieval of papers and help in preparing and formatting the report
Name:	Professor Aileen Clarke
Title:	Director of Warwick Evidence
Address:	Health Sciences Research Institute, Warwick Medical School, University of
	Warwick, Coventry CV4 7AL
Tel:	02476 150189
Email:	Aileen.Clarke@warwick.ac.uk
Contribution:	Co-ordinate review process, protocol development, data analysis, synthesis of
	findings and report writing

10.1 Methodological advisors

Professor Peter Elwood, Honorary Professor of Epidemiology, University of Cardiff Ms Sarah Morrow, Green Templeton College, University of Oxford Contribution of methodological advisor: previous experience of modelling in this area, multistate models, general evidence synthesis, statistics issues in health economic modelling, application of statistical methods to cardiothoracic medicine and surgery.

10.2 Clinical Advisors

Professor Martin Underwood, University of Warwick Dr Saverio Stranges, University of Warwick Dr Wendy Gregory, Clinical Consultant Gastroenterologist

Contribution of clinical advisors: protocol development, help interpret data, provide a methodological, policy and clinical perspective on data and review development of background information and clinical effectiveness and review of report drafts.

11. References

- 1. Baigent C, Blackwell L, Collins R, Emberson J, Godwin J, Peto R *et al.* Aspirin in the primary and secondary prevention of vascular disease: collaborative meta-analysis of individual participant data from randomised trials. *Lancet* 2009;373:1849-60.
- 2. Antithrombotic TC. Collaborative meta-analysis of randomised trials of antiplatelet therapy for prevention of death, myocardial infarction, and stroke in high risk patients.[Erratum appears in BMJ 2002 Jan 19;324(7330):141]. *BMJ* 2002;324:71-86.
- 3. Lanas A, Wu P, Medin J, Mills EJ. Low Doses of Acetylsalicylic Acid Increase Risk of Gastrointestinal Bleeding in a Meta-Analysis. *Clinical Gastroenterology and Hepatology* 2011;9:762-8.e6.
- 4. Lanas A. Gastrointestinal bleeding associated with low-dose aspirin use: relevance and management in clinical practice. *Expert Opin Drug Saf* 2011;10:45-54.
- 5. Jenkins C, Costello J, Hodge L. Systematic review of prevalence of aspirin induced asthma and its implications for clinical practice. *British Medical Journal* 2004;328:434-7.
- The Dutch TIA Trial Study Group. A comparison of two doses of aspirin (30 mg vs. 283 mg a day) in patients after a transient ischemic attack or minor ischemic stroke. N Engl J Med 1991;325:1261-6.
- 7. Sostres C, Lanas A. Gastrointestinal effects of aspirin. *Nat Rev Gastroenterol Hepatol* 2011;8:385-94.
- 8. Biondi-Zoccai GG, Lotrionte M, Agostoni P, Abbate A, Fusaro M, Burzotta F *et al.* A systematic review and meta-analysis on the hazards of discontinuing ir not adhering to aspirin among 50.279 patients at risk for coronary artery disease. *Eur Heart J* 2006;27:2667-74.
- 9. Pearson TA, Bazzarre TL, Daniels SR, Fair JM, Fortmann SP, Franklin BA *et al.* American Heart Association guide for improving cardiovascular health at the community level: a statement for public health practitioners, healthcare providers, and health policy makers from the American Heart Association Expert Panel on Population and Prevention Science. *Circulation* 2003;107:645-51.
- 10. Aspirin for the prevention of cardiovascular disease: U.S. Preventive Services Task Force recommendation statement. *Ann Intern Med* 2009;150:396-404.
- 11. Rothwell PM, Fowkes FG, Belch JF, Ogawa H, Warlow CP, Meade TW. Effect of daily aspirin on long-term risk of death due to cancer: analysis of individual patient data from randomised trials. *Lancet* 2011;377:31-41.
- 12. Chan AT, Cook NR. Are we ready to recommend aspirin for cancer prevention? *Lancet* 2012;379:1569-71.
- 13. Kurth T. Aspirin and cancer prevention. BMJ 2012;344:e2480.

- 14. Bartolucci AA, Tendera M, Howard G. Meta-analysis of multiple primary prevention trials of cardiovascular events using aspirin.[Erratum appears in Am J Cardiol. 2011 Aug 15;108(4):615]. *American Journal of Cardiology* 2011;107:1796-801.
- 15. Raju N, Sobieraj-Teague M, Hirsh J, O'Donnell M, Eikelboom J. Effect of aspirin on mortality in the primary prevention of cardiovascular disease. *Am J Med* 2011;124:621-9.
- 16. Berger JS, Lala A, Krantz MJ, Baker GS, Hiatt WR. Aspirin for the prevention of cardiovascular events in patients without clinical cardiovascular disease: A meta-analysis of randomized trials. *American Heart Journal* 2011;162:115-24.e2.
- 17. Belch J, MacCuish A, Campbell I, Cobbe S, Taylor R, Prescott R *et al.* The prevention of progression of arterial disease and diabetes (POPADAD) trial: factorial randomised placebo controlled trial of aspirin and antioxidants in patients with diabetes and asymptomatic peripheral arterial disease. *BMJ* 2008;337:a1840.
- 18. de GG. Low-dose aspirin and vitamin E in people at cardiovascular risk: a randomised trial in general practice. Collaborative Group of the Primary Prevention Project. *Lancet* 2001;357:89-95.
- 19. Fowkes FG, Price JF, Stewart MC, Butcher I, Leng GC, Pell AC *et al.* Aspirin for prevention of cardiovascular events in a general population screened for a low ankle brachial index: a randomized controlled trial. *JAMA* 2010;303:841-8.
- 20. Hansson L, Zanchetti A, Carruthers SG, Dahlof B, Elmfeldt D, Julius S *et al.* Effects of intensive blood-pressure lowering and low-dose aspirin in patients with hypertension: principal results of the Hypertension Optimal Treatment (HOT) randomised trial. HOT Study Group. *Lancet* 1998;351:1755-62.
- 21. Ogawa H, Nakayama M, Morimoto T, Uemura S, Kanauchi M, Doi N *et al.* Low-dose aspirin for primary prevention of atherosclerotic events in patients with type 2 diabetes: A randomized controlled trial. *Journal of the American Medical Association* 2008;30:2134-41.
- Peto R, Gray R, Collins R, Wheatley K, Hennekens C, Jamrozik K *et al.* Randomised trial of prophylactic daily aspirin in British male doctors. *Br Med J (Clin Res Ed)* 1988;296:313-6.
- 23. Ridker PM, Cook NR, Lee IM, Gordon D, Gaziano JM, Manson JE *et al.* A randomized trial of low-dose aspirin in the primary prevention of cardiovascular disease in women. *N Engl J Med* 2005;352:1293-304.
- 24. Final report on the aspirin component of the ongoing Physicians' Health Study. Steering Committee of the Physicians' Health Study Research Group. *N Engl J Med* 1989;321:129-35.
- 25. Thrombosis prevention trial: randomised trial of low-intensity oral anticoagulation with warfarin and low-dose aspirin in the primary prevention of ischaemic heart disease in men at increased risk. The Medical Research Council's General Practice Research Framework. *Lancet* 1998;351:233-41.

- 26. Rothwell PM, Wilson M, Price JF, Belch JF, Meade TW, Mehta Z. Effect of daily aspirin on risk of cancer metastasis: a study of incident cancers during randomised controlled trials. *Lancet* 2012;379:1591-601.
- 27. NHS Centre for Reviews and Dissemination. Undertaking systematic reviews of research on effectiveness: CRD guidelines for those carrying out or commissioning reviews. CRD Report 4. York: NHS Centre for Reviews and Dissemination, University of York. 1999.
- 28. Higgins J, Green S. Cochrane Handbook for Systematic Reviews of Interventions Version 5.1.0 [updated March 2011]. The Cochrane Collaboration, 2011. 2011. URL: www.cochrane-handbook.org (accessed 5 November 2011).
- 29. Fletcher EH, Johnston DE, Fisher CR, Koerner RJ, Newton JL, Gray CS. Systematic review: Helicobacter pylori and the risk of upper gastrointestinal bleeding risk in patients taking aspirin. *Alimentary Pharmacology and Therapeutics* 2010;32:831-9.
- McQuaid KR, Laine L. Systematic Review and Meta-analysis of Adverse Events of Lowdose Aspirin and Clopidogrel in Randomized Controlled Trials. *American Journal of Medicine* 2006;119:624-38.
- 31. Butalia S, Leung AA, Ghali WA, Rabi DM. Aspirin effect on the incidence of major adverse cardiovascular events in patients with diabetes mellitus: a systematic review and meta-analysis. *Cardiovasc Diabetol* 2011;10:25.
- 32. Thoonsen H, Richard E, Bentham P, Gray R, van GN, De Haan RJ *et al.* Aspirin in Alzheimer's disease: increased risk of intracerebral hemorrhage: cause for concern? *Stroke* 2010;41:2690-2.

12. Appendices

Appendix 1

Scoping search November 2011

Warwick Evidence carried out an overview of current relevant UK research related to potential harms from aspirin given in low dose (<300mgs) for any indication. We conducted a scoping search in November 2011 on harms of aspirin given in low doses and contacted UK and international expert in the field. The aim of the scoping searches was to present a short overview of the current status of policy and research in the UK and internationally concerning the potential harms from aspirin given in low dose (<300mgs) for any indication. This has informed the development of the current statues.

The following databases were searched: Medline (1948 to November 2011), Embase (1974 to November 2011), Cochrane (all sections), HTA (www.HTA.ac.uk), DARE were searched (until November 2011). No language filters were applied. Full search strategies are available on request from the authors. RCT and SR filters were applied to Medline, Embase as detailed in the search strategies. Combined searches produced 3064 references; de-duplicating the database resulted in a final set of 2981 references. Economics searches were undertaken in Medline (1948 to December 2011), EMBASE (1974 to December 2011) and NHS-EED. A search of the Current Controlled Trials Database (http://www.controlled-trials.com/mrct/) produced 629 results, of which, 44 were considered to be potentially relevant.

Five^{3,29-32} reviews were identified on the adverse events of low dose aspirin. The most recent review³ entitled "Low Doses of Acetylsalicylic Acid (ASA) Increase Risk of Gastrointestinal Bleeding in a Meta-Analysis" centred exclusively on risk of GI bleeding related to low dose aspirin (75-325 mg/d). The review included any randomised controlled studies that evaluated low-dose ASA, alone or in combination with anticoagulant, clopidogrel or proton pump inhibitors (PPIs). A total of 61 trials were included in the review. Thirty-five RCTs included analysis of ASA alone, and three RCTs included analysis of ASA plus proton pump inhibitors. The study reported all-cause mortality, fatal bleeding, and fatal GI bleeding, major bleeding, any bleeds (including cerebral bleed) and dyspepsia as their outcome.

Economic evidence was limited in comparison to clinical and public health evidence in this area. We did not identify any comprehensive reviews of cost or cost effectiveness on the topic and therefore a further analysis of cost-effectiveness or primary economic research will not be undertaken within the current work.

Appendix 2. Search strategy for Medline via OVID interface

Searched on 19/09/2012

1	exp *Aspirin/
2 3	(aspirin or acetylsalicyl* or "acetyl-salicyl*" or "acetyl salicyl*").tw. 1 or 2
4	(prevent* or prophyla*).tw.
5	exp Primary Prevention/
6	4 or 5
7	randomized controlled trial.pt.
8	(random* or controlled trial* or clinical trial* or rct).tw.
9	meta-analysis.pt.
10	("meta-analysis" or "meta analysis" or metaanalysis or "systematic review*").tw.
11	7 or 8 or 9 or 10
12	3 and 6 and 11
13	limit 12 to (english language and humans)

14 limit 13 to yr="2008 -Current"

Appendix 3. Data extraction form

a) Data extraction form for systematic reviews

Name of the reviewer:

Study details
Study ID (Ref man):
First author surname:
Year of publication:
Country:
Funding:
Aim of the study:
Methods
Databases searched:
Last date of search:
Inclusion criteria:
Participants:
Interventions:
Comparators:
Outcome measures:
Types of studies included:
Quality assessment criteria used:
Application of methods:
Methods of analysis:
1. narrative, 2. meta-analysis, 3. indirect comparison, 4. others
Results
Quantity and quality of included studies:
Treatment effect:
Economic evaluation:
Conclusions:
Implications of the review:
Methodological comments
Search strategy:

Participants:

Inclusion/exclusion criteria:

Quality assessment of studies:

Method of synthesis:

General comment

Generalisability:

Funding:

Authors conclusion

Reviewer's conclusion

b) Data extraction form for studies for primary prevention of cardiovascular events or cancers

Name of the reviewer:

Study details		
Study ID (Ref man):		
First author surname:		
Year of publication:		
Country:		
Study design:		
Study setting:		
Number of centres:		
Duration of study:		
Follow up period:		
Funding:		
Aim of the study:		
Participants		
Total number of participants:		
Sample attrition/drop out:		
Inclusion criteria:		
Exclusion criteria:		
Characteristics of participants:		
Mean age:		
Mean sex:		
Race:		
Date of diagnosis:		
Diagnosis:		
Diabetes (%):		
Smokers (%):		
Site/type of cancer to be prevented:		
Annual risk of cardiovascular events (%):		
Intervention		
Indication for treatment:		

Aspirin dose:			
Any comparison			
Duration of treatment:			
Compliance:			
Other interventions used:			
Outcomes			
Primary outcomes:			
Secondary outcomes:			
Method of assessing outcomes:			
Timing of assessment:			
Study end point:			
Survival analysis: Yes/No			
Mortality: Yes/No			
Adverse event: Yes/No			
Health related quality of life: Yes/No)		
Length of follow up:			
Number of participants	Intervention	Comparator	
Screened			
Screened Randomised/Included			
Screened Randomised/Included Excluded			
Screened Randomised/Included Excluded Missing participants			
Screened Randomised/Included Excluded Missing participants Withdrawals			
Screened Randomised/Included Excluded Missing participants Withdrawals Patient's baseline characteristics	Intervention	Comparator	
Screened Randomised/Included Excluded Missing participants Withdrawals Patient's baseline characteristics Insert baseline characteristics table h	Intervention ere	Comparator	
Screened Randomised/Included Excluded Missing participants Withdrawals Patient's baseline characteristics Insert baseline characteristics table h Survival data	Intervention ere Intervention	Comparator Comparator	
Screened Randomised/Included Excluded Missing participants Withdrawals Patient's baseline characteristics Insert baseline characteristics table h Survival data Actuarial survival	Intervention ere Intervention	Comparator Comparator	
Screened Randomised/Included Excluded Missing participants Withdrawals Patient's baseline characteristics Insert baseline characteristics table h Survival data Actuarial survival Overall survival	Intervention ere Intervention	Comparator Comparator	
Screened Randomised/Included Excluded Missing participants Withdrawals Patient's baseline characteristics Insert baseline characteristics table h Survival data Actuarial survival Overall survival Kaplan-Meier estimates	Intervention ere Intervention	Comparator Comparator	
Screened Randomised/Included Excluded Missing participants Withdrawals Patient's baseline characteristics Insert baseline characteristics table h Survival data Actuarial survival Overall survival Kaplan-Meier estimates Survival by era (at 5 year intervals)	Intervention ere Intervention	Comparator Comparator	
Screened Randomised/Included Excluded Missing participants Withdrawals Patient's baseline characteristics Insert baseline characteristics table h Survival data Actuarial survival Overall survival Kaplan-Meier estimates Survival by era (at 5 year intervals) Adverse events	Intervention Intervention Intervention Intervention Intervention Intervention Intervention	Comparator Comparator Comparator Comparator Comparator Comparator	
Screened Randomised/Included Excluded Missing participants Withdrawals Patient's baseline characteristics Insert baseline characteristics table h Survival data Actuarial survival Overall survival Overall survival Kaplan-Meier estimates Survival by era (at 5 year intervals) Adverse events Bleeding/haemorrhagic end points	Intervention ere Intervention Intervention Intervention Intervention Intervention	Comparator Comparator Comparator Comparator Comparator	
ScreenedRandomised/IncludedExcludedMissing participantsWithdrawalsPatient's baseline characteristicsInsert baseline characteristics table hSurvival dataActuarial survivalOverall survivalKaplan-Meier estimatesSurvival by era (at 5 year intervals)Adverse eventsBleeding/haemorrhagic end pointsStroke	Intervention ere Intervention Intervention Intervention Intervention Intervention	Comparator Comparator Comparator Comparator Comparator	

Peptic ulcer			
Rashes			
Wheezing/asthma			
• Episodes			
• Mortality			
Quality of life	Intervention	Comparator	
Authors conclusion			
Reviewer's conclusion			

Appendix 4. Quality assessment forms

Quality assessment criteria for systematic reviews: Based on NHS Centre for Reviews and Dissemination (CRD)²⁷

Question	Score
1. Are any inclusion/exclusion criteria reported to the primary studies which address	Yes or No
the review question?	
2. Is there evidence of a substantial effort to search for all relevant research?	Yes or No
3. Is the validity of included studies adequately assessed?	Yes or No
4. Is sufficient detail of the individual studies presented?	Yes or No
5. Are the primary studies summarised appropriately?	Yes or No

Quality assessment criteria for RCTs: *Based on the Cochrane Risk of Bias tool*²⁸

Question	Rating
1. Adequate sequence generation	
2. Adequate allocation concealment	
3. Blinding (especially outcome assessment)	
4. Incomplete outcome data addressed	
5. Free of selective reporting	
6. Free of other bias (e.g. similarity at baseline, power assessment, conflict of interest)	

Rating (by criteria fulfilled, i.e. 'yes' response): 0 to 2 low quality, 3 to 4 medium quality, 5 to 6 high quality